Author’s response to reviews

Title: Design and implementation of a mobile health electronic data capture platform that functions in fully-disconnected settings: a pilot study in rural Liberia

Authors:

Avi Kenny (avikenny@uw.edu)
Nicholas Gordon (ngordon@lastmilehealth.org)
Jordan Downey (jdowney@lastmilehealth.org)
Owen Eddins (oeddins@lastmilehealth.org)
Kathleen Buchholz (kbuchholz@lastmilehealth.org)
Alvin Menyon (amenyon@lastmilehealth.org)
William Mansah (bill.mansah@yahoo.com)

Version: 1 Date: 16 Dec 2019

Author’s response to reviews:

See attachments for response letter to editor and reviewers.